Author's response to reviews

Title: Household cost-benefit equations and equity in immunization: a randomised cluster controlled trial of knowledge translation for sustainable universal childhood immunization in south Pakistan [ISRCTN12421731]

Authors:

Neil Andersson (Neil@ciet.org)
Anne Cockcroft (CIETeurope@compuserve.com)
Noor Ansari (nansari@ciet.org)
Khalid Omer (komer@super.net.pk)
Joe Losos (jlosos@uottawa.ca)
Beverley Shea (bshea@ciet.org)

Version: 2 Date: 24 June 2005

Author's response to reviews:

MS: 9168829546919712

Responses to Review of 'Household cost-benefit equations and equity in immunization: a randomised cluster controlled trial of knowledge translation for sustainable Universal Childhood Immunization in south Pakistan [ISRCTN12421731]'.

COMMENT:

Please ensure the title and abstract are identical in the manuscript and in the submission system.

RESPONSE:

This has been reviewed and corrected, and the title and abstract are now both identical in the manuscript and in the submission system.

COMMENT:

Please remove titles from the list of authors.

RESPONSE:

This has been performed.

COMMENT:

Please use sentence case in the title, both in the manuscript file and the submission system (i.e. remove all unnecessary capitals).

RESPONSE:

All unnecessary capitals have been removed from the title in both the manuscript file and the submission system.

COMMENT:

Each affiliation should be written in full and be provided in the following format: department, institute, city, country. Please do not use full capitals for country names, and please do not include full-stops in abbreviated country names. We also ask you to link each author to their corresponding affiliation with the use of superscript numbers.

RESPONSE:
Each affiliation has been written in full in the requested format, and country names do not include full capitals or full-stops for abbreviations. Also, each author has been linked to their corresponding affiliation in the requested format.

COMMENT:

References should be cited in the text using consecutive numbers in square brackets.

RESPONSE:

Square brackets have been added to the in-text citations. Also, the reference numbers provided in the text have been reviewed and are in consecutive order.

COMMENT:

In the manuscript text, please place the reference before the full stop.

RESPONSE:

This has been rectified.

COMMENT:

For the authors' contributions we suggest the following kind of format (please use initials to refer to each author's contribution): AB carried out the molecular genetic studies, participated in the sequence alignment and drafted the manuscript. JY carried out the immunoassays. MT participated in the sequence alignment. ES participated in the design of the study and performed the statistical analysis. FG conceived of the study, and participated in its design and coordination. All authors read and approved the final manuscript.

RESPONSE:

The authors' contributions have been rewritten in the requested format, and initials were used to refer to the authors.

COMMENT:

Please acknowledge anyone who contributed towards the study by making substantial contributions to conception, design, acquisition of data, or analysis and interpretation of data, or who was involved in drafting the manuscript or revising it critically for important intellectual content, but who does not meet the criteria for authorship. Please also, include their source(s) of funding. Please also acknowledge anyone who contributed materials essential for the study.

Authors should obtain permission to acknowledge from all those mentioned in the Acknowledgements. Please list the source(s) of funding for the study, for each author, and for the manuscript preparation in the acknowledgements section. Authors must describe the role of the funding body, if any, in study design; in the collection, analysis, and interpretation of data; in the writing of the manuscript; and in the decision to submit the manuscript for publication.

RESPONSE:

There are no additional persons that have contributed towards the study other than the authors. The study, the authors, and the manuscript preparation are all funded by the Canadian International Development Agency (CIDA) in collaboration with the International Development Research Centre (IDRC), the Canadian Institutes of Health Research (CIHR) and Health Canada (HC). (File: 102172-007). The funding body had no role in study design; in the collection, analysis, and interpretation of data; in the writing of the manuscript; and in the decision to submit the manuscript for publication.

COMMENT:

The reference list should be formatted in the journal's style to enable us to create links to article abstracts on PubMed.
RESPONSE:

The reference list has been edited and is now formatted in the journal's style.

COMMENT:

Please amend the 'Competing interests' section. If there are none to declare, please write 'The authors declare that they have no competing interests'.

RESPONSE:

There are no competing interests to declare, and the requested statement has been inserted into the 'Competing interests' section.

We appreciate your constructive comments, and would like to thank you for your review.